Author’s response to reviews

Title: Stressors and resources related to academic studies and improvements suggested by medical students: a qualitative study

Authors:

Jeannette Weber (Jeannette.Weber@uni-duesseldorf.de)

Stefanie Skodda (StefanieSkodda@gmx.de)

Thomas Muth (mutht@uni-duesseldorf.de)

Peter Angerer (peter.angerer@uni-duesseldorf.de)

Adrian Loerbroks (Adrian.Loerbroks@uni-duesseldorf.de)

Version: 4 Date: 05 Aug 2019

Author’s response to reviews:

Dear Ms. Fagelson,

Dear Dr. Choi-Lundberg,

thank you very much for reassessment and recommendation of our manuscript “Stressors and resources related to academic studies and improvements suggested by medical students: a qualitative study”.

We have revised our manuscript according to your suggestions, including acceptance of all minor edits, provision of a list of abbreviations and checked whether our headings are in line with journal requirements. The final manuscript, figures and additional files are now uploaded as clean versions without any tracked changes.

Yours sincerely,

Peter Angerer, Jeannette Weber, Thomas Muth, Stefanie Skodda and Adrian Loerbroks